Author’s response to reviews

Title: A test of financial incentives to improve warfarin adherence

Authors:

Kevin G Volpp (volpp70@mail.med.upenn.edu)
George Loewenstein (gl20@andrew.cmu.edu)
Andrea B Troxel (atroxel@mail.med.upenn.edu)
Jalpa Doshi (jdoshi@mail.med.upenn.edu)
Maureen Price (maureenpprice@hotmail.com)
Mitchell Laskin (mitchell.laskin@uphs.upenn.edu)
Stephen E Kimmel (stevek@mail.med.upenn.edu)

Version: 4 Date: 21 November 2008

Author’s response to reviews:

1. Please separate emails from the author list and place them after the affiliations.

   This has been done.

2. Change Financial Competing Interests -> Competing interests

   Done

3. Author Contributions -> Authors’ contributions

   Done

4. Authors contributions- please ensure this is formatted into a paragraph rather than a list or in bullet points.

   Done

Section headings. Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT, Figure legends not Figure Legends.

Done

Please place the phrase Figure legends above the legends.

Done

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Done